Author's response to reviews

Title: Central corneal sensitivity after small incision lenticule extraction versus femtosecond laser-assisted LASIK for myopia: a meta-analysis of comparative studies

Authors:

Miao He (943602453@qq.com)
Wenyong Huang (andyhwyz@aliyun.com)
Xingwu Zhong (zhongxwu@mail.sysu.edu.cn)

Version: 3
Date: 2 October 2015

Author's response to reviews: see over
Dear Prof. Rada,

Thank you very much for your consideration for our manuscript entitled “Small-incision lenticule extraction versus femtosecond laser-assisted LASIK for myopia: a meta-analysis of comparative studies” (MS: 3103889271669457). Our point-by-point responses to the comments and suggestions are listed below.

We hope the revised version of the manuscript is acceptable for the publication in the journal.

Sincerely yours

Xingwu Zhong
Zhongshan Ophthalmic Center and Sate Key Laboratory of Ophthalmology, Sun Yat-sen University
54SXianlie Road, Guangzhou, China 510060
Email:zhongxwu@mail.sysu.edu.cn
Tel: 86-20-87330381
Fax: 86-20-87333190

Responses to the Editorial Requirements

1. Please include a list of keywords and List of abbreviations that were used in the manuscript.

Response: Thanks for your advices. The keywords and list of abbreviations were added in the revised manuscript.

2. Please ensure main section headings in the abstract and main text are as outlined in the journal author instructions

Response: Yes, we have rechecked it.

3. Please check the journal criteria for authorship and ensure all authors meet these criteria. Please use initials rather than full names in this section.

Response: The authorships were revised as your suggestions.